\begin{table}%t1
%\centering
\par
\caption{\label{tableau1}Basic properties of And~II taken from:
1 -- Schlegel et~al. (\cite{SchFD98}); 2 -- van den Bergh (\cite{vdBer00});
3 -- Amandroff \&{} da~Costa (\cite{ArmaD99}); 4 -- Mateo 
(\cite{Mateo98}); 5 -- C{\^o}t{\'e} et~al. (\cite{CotOC99}).}
\begin{tabular}{lccc}
\hline
\hline
               &     & And II                                      &Ref.\\
\hline
$\alpha_{2000}$$^*$&     & 1$^{\rm h} 16^{\rm m}27^{\rm s}$            &4\\
$\delta_{2000}$$^*$&     & $+33^{\circ}25^{\prime}42^{\prime\prime}$   &4\\
type           &     & dSph                                        &2\\
$D$            &[kpc]& 700                                         &3\\
($m{-}M$)$_0$&[mag]& 24.22                                       &3\\
$M_V$          &[mag]& $-$11.22                                    &4\\
${\rm [Fe/H]}$       &[dex]& $-$1.47                                     &5\\
$E_{V-I}$      &[mag]& 0.09                                        &1\\
\hline
\end{tabular}
\\
$^*$ The observation were done following coordinates given by NASA Extragalactic Database NED, resulting in an off-centered field.
\end{table}